Author's response to reviews

Title: Intestinal neuroendocrine tumor in a patient with pituitary adenoma. A case report and review of the current screening recommendations

Authors:

Cherif Boutros (cherifboutros@hotmail.com)
Diana Cheng-Robles (chengrdi@umdnj.edu)
Robert Goldenkranz (rgoldenkranz@sbhcs.com)

Version: 5 Date: 23 October 2007

Author's response to reviews:

The following corrections were performed according to you letter:

Author email addresses - Please give a different email address for each author, on the title page, beneath the corresponding author details.
>Author email addresses added

Authors' contributions - Please indicate that all authors read and approved the final manuscript.
>Statement added

Please put the statement of consent under a section entitled Acknowledgments.
>Consent statement placed under Acknowledgments

References - The reference list should contain all authors' names, if there are 30 or below. The term 'et al.' should not be used.
>All Authors name were included

Figures - There are six figure legends but only five uploaded figure files and only five figures referenced in the text.
>Figure legends corrected to a total of five.